Author’s response to reviews

Title: Frequency and risk factors for recurrent gestational diabetes mellitus in primiparous women: a case control study

Authors:

Yin-Yu Wang (wangyinyu2009@foxmail.com)
Ye Liu (liuye90s@126.com)
Cheng Li (centrino-li@hotmail.com)
Jing Lin (linjing1215@163.com)
Xin-Mei Liu (nanlilac@hotmail.com)
Jian-Zhong Sheng (shengjz@zju.edu.cn)
HEFENG HUANG (huanghefg@hotmail.com)

Version: 2 Date: 08 Aug 2018

Author’s response to reviews:

Dear editors, Each of the comments of the two Reviewers has been addressed, and a point-by-point response detailing the revisions is provided below. The modified parts have been highlighted in yellow in the revised manuscript. thank you very much!